Author’s response to reviews

Title: Complete biopsy-proven resolution of deposits in recurrent proliferative glomerulonephritis with monoclonal IgG deposits (PGNMIGD) following Rituximab treatment in renal allograft.

Authors:

Jon von Visger (jon.vonvisger@osumc.edu)

Gerardo Franco-Ahumada (gerardo.francoahumada@osumc.edu)

Uday Nori (uday.nori@osumc.edu)

Tibor Nadasdy (tibor.nadasdy@osumc.edu)

Anjali Satoskar (anjali.satoskar@osumc.edu)

Version: 1 Date: 01 Mar 2018

Author’s response to reviews:

1. The declarations have been added at the end of the manuscript. The patient has verbally consented to allow publication after removal of all identifying information. This is a retrospective study. It does not involve any prospective sample collection. But still, written consent is in process. To keep up with the time limit provided, we are submitting the manuscript. The written consent form will be submitted before any decision on the manuscript is made.

2. The author e-mails have been added.

3. Author contributions have been added under Declarations.

4. The key words have been added after the abstract.